Prediction and Description in Marxian Value Theory

By JONATHAN F. COGLIANO*

This paper reflects on recent work in Marxian value theory, with an emphasis on contributions by Foley and Shaikh, as well as those following their work. Sen's (1978) taxonomy of "predictive" and "descriptive" aspects of Marx's theory is adopted to appraise and discuss these recent contributions. Many have come to doubt Marx's theory of value as a predictive theory given the outcomes of the transformation problem debates, although recent work by Shaikh provides a forceful counter to this. Even if many remain skeptical of Marx's theory of value as a predictive theory, there are fruitful interpretations of Marx's theory as a descriptive theory, where the emphasis is on describing fundamental processes of capitalist economies like the allocation of social labor, capitalist competition, and exploitation. Recent work in Marxian value theory by Foley, Shaikh, and others has facets that are both predictive and descriptive. This short reflection highlights these and points to paths for future research.

JEL: B24, B51, D46

Keywords: Competition, labor theory of value, labor values, Marxian value theory, surplus value

I. Introduction

Duncan Foley and Anwar Shaikh have made fundamental contributions to Marxian value theory, and their work shapes much of the landscape of current debates in the field. These contributions include: the role of money in Marx's theory of value and the so-called 'New Interpretation' (NI) of Marx (Foley 1982, 1986, 2005); approaches to the infamous 'transformation problem' (Foley 1982, 1986, 2011*b*; Shaikh 1977, 1981, 1982, 1984, 1992, 2016); classical-Marxian competition and profit rate dynamics (Scharfenaker and Foley 2017; Shaikh 1978, 1979, 1980*b*, 2008, 2016); and applications of the labor theory of value for understanding the dynamics of capitalist economies (Basu and Foley 2013; Foley 2011*c*, 2013, 2016, 2018; Shaikh 1998, 2016; Shaikh and Tonak 1994), among other contributions to Marxian economics more broadly. This paper provides a brief reflection on select key contributions by Foley and Shaikh to Marxian value theory. Attention is also paid to contributions by others following the work of Foley and Shaikh, as well as those working in similar directions. Foley and Shaikh's many contributions in other areas of classical-Marxian economics, as well as other areas of economics in general, are beyond the current scope.

Marx's theory of value, or what is often referred to as the labor theory of value (LTV), has many critics, both in mainstream and heterodox circles. One source of criticism is the transformation problem and issues of consistency between the labor content of commodities—or *labor values*—and prices, see for instance Samuelson (1970, 1971, 1974) and Steedman (1977), among others. A primary problem for critics is that labor values and prices cannot be shown to be theoretically consistent, and as a result, labor values do not determine prices, or are at best unnecessary for analyzing prices. Marx's theory of value has also been charged as being overly, or purely, "metaphysical" and lacking in substance for economic analyses (Robinson 1964).

The contributions by Foley and Shaikh and others, discussed here provide counters to these criticisms. Some argue that properly conceptualized notions of labor values (and the LTV) do bear a close relationship with prices, and thus are relevant for understanding the dynamics of equilibrium and relative

^{*} Economics Department, University of Massachusetts Boston (jonathan.cogliano@umb.edu). The author owes a great deal to the intellectual influences of Duncan Foley and Anwar Shaikh, which have shaped this paper and will continue to shape the work of generations working in Marxian political economy. Thanks are owed to two anonymous referees for their helpful comments and suggestions. The usual disclaimer applies.

prices.¹ Others take a different approach and reconsider the role of labor values and the LTV—as well as other categories which can be grouped under the heading of "labor accounts"—in Marxian economics (Cogliano, Veneziani and Yoshihara 2022).² In light of the vast literature responding to critics of Marx, in one way or another, it is important to note that there is no singularly unique way to interpret Marx's original work, and there are multiple "logically consistent" ways to approach Marxian value theory (Mohum and Veneziani 2017). Thus, despite the criticisms, there remain different interpretations of Marx's theory of value that yield robust insights into capitalist economies.

The typical view, of both critics and proponents of Marx, is that the LTV is a *predictive* theory since labor values should predict, or determine, prices. The outcome of the transformation problem debates has led many to doubt the LTV as a predictive theory. However, Shaikh's (1984; 1998; 2016) work, along with others, provides a forceful response to this view. Relatedly, even if labor values do not predict prices there are other useful predictive aspects of the LTV, particularly establishing a relationship between exploitation and profits.³ Even if the LTV has usefulness as a predictive theory, "a strictly predictive interpretation of Marxian value theory is unnecessarily reductive" and needlessly limits the scope of analysis in Marxian economics (Mohun and Veneziani 2017, p.1415). Marx's theory of value may have limitations as a predictive theory, but there is an arguably fruitful approach to viewing Marx's theory of value as a *descriptive* theory. Thus, despite the critics, there are valid predictive and descriptive aspects of the LTV, and it is certainly not just metaphysical (Mohun and Veneziani 2017).

Sen (1978, p.175) identifies "predictive", "descriptive", and "normative" interpretations of Marx's theory that are "non-metaphysical".⁴ Predictive interpretations of the LTV tend to focus on Marx's theory as a "theory of prices", but there is additional "predictive value" of the LTV in analyzing class and exploitation (Sen 1978, pp.175, 182). While treating the LTV as a theory of prices can run up against the problems raised in the transformation debates, the descriptive interpretation of Marx's theory can be quite compelling.

In Sen's (1978, p.177) analysis, Marx's theory of value can be taken as describing "the *activity* of production [original emphasis]" and "participation in production as personal participation through labour". This interpretation draws on Dobb (1937, 1973) and his emphasis on the *human* focus of Marx's ideas, particularly the central role assigned to "human productive activity" (Dobb 1973, p.145). As Sen (1978, p.177) summarizes, "Dobb saw in the labour theory of value a description of production that is focused on human beings—giving a quantitative expression to human effort". This approach adopts a broader notion of the concept of labor and a wider scope for economic analysis than standard approaches focused on prices and efficiency (Dobb 1973, p.145). A primary benefit of the descriptive interpretation of Marx's theory of value is that it highlights the central role of human labor effort in describing fundamental processes of capitalist economies like the activity of production, the allocation of social labor, capitalist competition, and exploitation (Cogliano et al. 2018; Cogliano 2023*a*).

These topics are of central concern in Marxian economics and the LTV can serve as a robust foundation for analyzing them. This paper reflects on the aforementioned contributions by Foley, Shaikh, and others with a focus on the aspects of these that align with treating Marx's theory of value as a descriptive theory.⁵ This is not to downplay the predictive aspects of these contributions, but to emphasize what standard critiques of Marx miss. The aim here is to highlight the explanatory power

¹See, for instance, Chilcote (2004), Cockshott and Cottrell (1997), Fröhlich (2013), Işikara and Mokre (2022), Ochoa (1989), Shaikh (1984, 1998, 2016), and Tsoulfidis and Maniatis (2002).

²See, for instance, Basu and Foley (2013), Duménil (1980, 1984), Cogliano (2018, 2023*a*,*b*), Cogliano et al. (2018), Cogliano, Veneziani and Yoshihara (2022), Foley (1982, 2000, 2011a,b,c, 2013, 2016, 2018), Flaschel, Franke and Veneziani (2013), Flaschel, Frölich and Veneziani (2013), Mohun (1994, 1996, 2004, 2005, 2006, 2009, 2013, 2014), Moseley (1986, 1988, 1990, 1991, 1993, 1994, 1998, 2000), and Yoshihara (2021), among others.

³See Veneziani and Yoshihara (2015, 2017*a*,*b*) and Yoshihara and Veneziani (2018) for recent axiomatic studies of the relationship between exploitation and profits, which are related to concepts from Duménil (1980) and Foley (1982).

 $^{^{4}}$ The normative interpretations of the LTV identified by Sen (1978) are beyond the scope of this paper. See Mohun and Veneziani (2017) and Cogliano, Veneziani and Yoshihara (2022) for discussion of the normative relevance of the LTV and labor values in measuring exploitation and inequality. See also Cogliano, Veneziani and Yoshihara (2016, 2019, 2021) for recent examples along these lines.

 $^{{}^{5}}$ It is worth noting that there are other areas of work by Shaikh that fit with the notion of descriptive theory used here. For instance, Shaikh's (1974; 1980*a*) work on aggregate production functions reveals their shortcomings for understanding distribution, and demonstrates how aggregate production functions provide poor descriptions of the labor process and other fundamental processes of capitalism. However, these contributions are beyond the scope of this paper.

of Marx's theory of value as a descriptive theory, and to briefly chart some paths forward for research. This is not a thorough or detailed survey, nor is the aim to rehash all the points of debate in Marxian value theory. This paper is meant to provide 'food for thought' for future research, and uses Foley and Shaikh's contributions as a foundation for discussion.

The remainder of the paper proceeds as follows. Section II presents the main issues raised by the transformation problem to establish the reference point for debates around Marx's theory of value while section III discusses aspects of Marx's method that are often overlooked in the transformation problem debates, as well as some related contributions by Foley and Shaikh that respond to critics of the LTV. Sections IV-VI discusses capitalist competition; section V discusses labor mobility and the allocation of social labor; and section VI discusses recent work advocating the useful role of labor accounts in analyzing capitalist economies. Section VII concludes.

II. Labor Values, Prices, and Skepticism

The main issues raised by the transformation problem can be easily seen through the standard "dualsystem" or "dualist" approach to labor values and prices (Foley 2000; Mohun and Veneziani 2017). The dual-system approach is often attributed to Morishima (1973) and Steedman (1977), among others, and it has become a reference point for much of the recent discourse in Marxian value theory. Both positive and negative perspectives on Marx's theory of value are often influenced, in some way, by the dualsystem framework. The brief exposition here helps to clearly frame the sources of skepticism toward Marxian value theory, as well as the responses to the transformation debates that argue for the empirical and theoretical relevance of the LTV.⁶ The presentation here is also helpful for defining concepts used throughout the remainder of this paper, and it frames the discussion of prediction and description in recent contributions in Marxian value theory.

Assume a linear circulating capital economy composed only of n productive industries, where each industry produces a single commodity and operates a single activity in production (i.e. there is no joint production and no multiple activities). Let each industry be denoted by j = 1, ..., n and let prices be denoted by the $1 \times n$ row vector \mathbf{p} . Let \mathbf{A} represent the $n \times n$ matrix of capital requirements for production (or non-labor inputs), with each element a_{ij} being the quantity of commodity i needed to produce a single unit of commodity j and column \mathbf{A}_j denoting the capital requirements of industry j. Let \mathbf{L} be the $1 \times n$ vector of direct labor requirements with element l_j . Let \mathbf{b} be the $n \times 1$ subsistence bundle of workers, and let w and r, respectively, denote profit and wage rates. The gross output (or activities) vector for the economy is denoted by the $n \times 1$ vector \mathbf{x} , and the vector of net output is $\mathbf{y} = (\mathbf{I} - \mathbf{A})\mathbf{x}$, where \mathbf{I} denotes the n-dimensional identity matrix. Lastly, let \mathbf{v} denote the $1 \times n$ vector of labor values with element v_j . Under the dual-system, labor values are defined as $\mathbf{v} = \mathbf{vA} + \mathbf{L} = \mathbf{L} (\mathbf{I} - \mathbf{A})^{-1}$.

The transformation problem debates revolve around whether or not three relationships known as 'Marx's identities' can be shown to hold (Foley 2000). These identities deal with aggregate relationships across the *n*-commodity economy described above. They are: (i) total labor value is proportional to total price; (ii) total surplus value is proportional to total profit; and (iii) the profit rate calculated in labor values is equal to the profit rate calculated in price terms. These relationships cannot hold in the dual-system framework, except under restrictive scenarios.

First, whether one uses gross output or net output to determine total value and total price, although it is typical to use gross output, (i) cannot hold in general. Using gross output, total value is \mathbf{vx} , and using net output, it is $\mathbf{vy} = \mathbf{L} (\mathbf{I} - \mathbf{A})^{-1} (\mathbf{I} - \mathbf{A}) \mathbf{x} = \mathbf{Lx}$. Similarly, total price is \mathbf{px} or $\mathbf{py} = \mathbf{p} (\mathbf{I} - \mathbf{A}) \mathbf{x}$ using gross or net output, respectively. For any gross output vector $\mathbf{x} > \mathbf{0}$, \mathbf{vx} and \mathbf{px} , or \mathbf{vy} and \mathbf{py} , are proportional if $\mathbf{p} = \phi \mathbf{v}$, where ϕ is some positive scalar. If prices are equilibrium prices, or prices of production, whether workers are paid after production, $\mathbf{p} = (1 + r)\mathbf{pA} + w\mathbf{L}$, or prior to production, $\mathbf{p} = (1 + r)(\mathbf{pA} + w\mathbf{L})$, for any r > 0, $\mathbf{p} \neq \phi \mathbf{v}$ as long as \mathbf{A}_j/l_j are different for some $j = 1, \ldots, n$. Stated differently, equilibrium prices are proportional to labor values *if and only if* the profit rate is zero or organic compositions of capital across industries are uniform. A zero profit rate and

 $^{^{6}}$ See Desai (1988), Foley (2000), and Mohun and Veneziani (2017) for surveys of the transformation problem debates and developments in Marxian value theory.

uniform organic compositions of capital are somewhat restrictive—and arguably unrealistic—scenarios, thus proportionality between prices and labor values does not hold in general. Thus, \mathbf{vx} and \mathbf{px} will not be proportional and neither will \mathbf{vy} and \mathbf{py} under the conditions outlined here.

Similarly, for (ii), the lack of proportionality between labor values and prices means that total surplus value and total profit will not be proportional. Formally, total surplus value is $\mathbf{v} (\mathbf{I} - \mathbf{A} - \mathbf{bL}) \mathbf{x}$ and total profit is $\mathbf{p} (\mathbf{I} - \mathbf{A}) \mathbf{x} - w \mathbf{L} \mathbf{x}$, and these are not proportional so long as $\mathbf{p} \neq \phi \mathbf{v}$.⁷ The same applies for (iii) and equality of the 'value profit rate' and 'price profit rate'. The value profit rate is $\frac{\mathbf{v}(\mathbf{I} - \mathbf{A} - \mathbf{bL})\mathbf{x}}{\mathbf{v}(\mathbf{A} + \mathbf{bL})\mathbf{x}}$ and the price profit rate is $\frac{\mathbf{p}(\mathbf{I} - \mathbf{A})\mathbf{x} - w \mathbf{L} \mathbf{x}}{\mathbf{p}\mathbf{A}\mathbf{x} + w \mathbf{L}\mathbf{x}}$, and these cannot be equal if $\mathbf{p} \neq \phi \mathbf{v}$.⁸

There are limited cases where Marx's identities can hold under the dual-system even if $\mathbf{p} \neq \phi \mathbf{v}$. As Morishima (1974) shows, if the gross output vector (or aggregator) is selected from along the so-called 'von Neumann ray', i.e. "the long-run equilibrium balanced-growth output vector", (i)-(iii) can hold.⁹ Assuming the wage is equal to subsistence, $w = \mathbf{pb}$, and that wages are paid prior to production, the outlays required for production can be written as the matrix $\mathbf{M} = \mathbf{A} + \mathbf{bL}$. Let λ denote the largest positive eigenvalue of \mathbf{M} , and let $\bar{\mathbf{x}}$ denote the eigenvector associated with λ . Letting the subscript tdenote time periods for production, prices can be written as $\mathbf{p}_t = (1 + \bar{r})\mathbf{p}_{t-1}\mathbf{M}$, where $(1 + \bar{r}) = 1/\lambda$. Taking \bar{r} as the equilibrium profit rate for long-period prices, if initial prices are equal—or at least very close—to labor values, $\mathbf{p}_0 = \mathbf{v}$, then the price vector will eventually converge to one where Marx's identities hold, provided $\bar{\mathbf{x}}$ is the gross output vector. Letting $\bar{\mathbf{p}} = \lim_{t\to\infty} \mathbf{p}_t$ where $\mathbf{p}_0 = \mathbf{v}$, total value will be equal to total price, $\mathbf{v}\bar{\mathbf{x}} = \bar{\mathbf{p}}\bar{\mathbf{x}}$. Similarly, total surplus value will be equal to total profit, $\mathbf{v}(\mathbf{I} - \mathbf{M})\bar{\mathbf{x}} = \bar{\mathbf{p}}(\mathbf{I} - \mathbf{M})\bar{\mathbf{x}}$, and the value and price profit rates will be equal, $\frac{\mathbf{v}(\mathbf{I}-\mathbf{M})\bar{\mathbf{x}}}{\bar{p}M\bar{\mathbf{x}}} = \frac{\bar{\mathbf{p}}(\mathbf{I}-\mathbf{M})\bar{\mathbf{x}}}{\bar{p}M\bar{\mathbf{x}}}$. Thus, under Morishima's (1974) assumptions, (i)-(iii) can hold.

Morishima's (1974) results depend on assumptions about wages, initial prices, and the gross output vector.¹⁰ The assumptions necessary for these results to hold could be seen as restrictive and Marx's identities will still fail to hold in more general circumstances. For the specific vector $\bar{\mathbf{x}}$, lack of proportionality between prices and labor values does not pose an issue. However, for *any* gross output vector $\mathbf{x} > \mathbf{0}$, the lack of proportionality between prices and labor values and prices, and problems maintaining Marx's identities, are a primary source of doubts about the LTV, particularly as a predictive theory.

These doubts have also led some to disregard the concept of labor values and other Marxian concepts like surplus value, and in general they have fueled questions about the LTV as a relevant basis for the Marxian notion of exploitation—with some going as far as concluding "that the relationship between the labor theory of value and the concept of exploitation is one of mutual irrelevance" (Cohen 1979, p.338). The rejection of labor accounts by many in the mainstream, as well as within heterodox circles, however, misses the usefulness of these concepts as part of a descriptive theory of capitalist economies.

As Foley (2000) summarizes, the outcome of the transformation problem debates in the second-half of the 20th century led to different responses by those who maintained the relevance of Marxian value theory, or at least some related Marxian concepts. One set of responses generally accepts the outcomes of the transformation debates and shows that a theory of exploitation, as well as a capitalist class structure, can be built on alternatives bases like differential ownership of wealth (or productive assets), without the need for consistency between labor values and prices—see for instance Roemer (1982) and others in what is sometimes called the 'Analytical Marxist' tradition.¹¹ However, other responses counter the generally received view on the transformation debates and establish both the theoretical and empirical relevance of labor accounts and Marxian value theory. Many of these contributions also highlight the human aspects of Marx's theory of value and its power as a descriptive theory.

⁷The lack of proportionality between total surplus value and total profit when prices are not proportional to labor values holds even if the wage is equal to the price of the subsistence bundle, i.e. $w = \mathbf{pb}$.

 $^{^{8}}$ Marx's identities also do not hold under other definitions of prices as long as prices are not proportional to labor values. See Wright (2014, 2019) for an example of an approach that redefines labor values such that proportionality and Marx's identities can hold.

⁹Thanks are owed to an anonymous referee for pointing out the nuanced importance of Morishima's (1974) main results for the transformation problem debates.

 $^{^{10}}$ It is worth noting that unless prices begin at or very close to labor values they will not converge to equilibrium prices that allow Marx's identities to hold.

¹¹See Veneziani (2012) for a survey of both analytical and "rational choice Marxism".

III. Circulation, Money, and the Human Element

The conclusions of the dual-system approach are mathematically true. However, some may argue that the dual-system runs the risk of casting Marx's ideas in a "flat conceptual space", which overlooks important aspects of Marx's theory of value (Shaikh 1982, p.68). The rejection of labor values and labor accounts leads to a heavy focus on the properties of prices, and inherently misses descriptive aspects of Marx's ideas. Inconsistencies between prices and labor values withstanding, the "descriptive aspect" of Marx's theory of value "remains the capturing of the human dimension of production and distribution" (Sen 1978, p.183). Prices alone "lack this feature of describing personal participation" and the human dimension of Marx's theory (Sen 1978, p.179).

As Shaikh (1981, 1982) points out, human aspects like the labor process and production process fall into the backdrop in the dual-system, and the important "qualitative" features of Marx's theory of value can become lost. Similarly, as Foley (2016) argues, an overemphasis on prices cannot allow one to make a clear connection between the expenditure of human labor effort and value creation. Approaches adhering too closely to the mathematical properties of systems of prices and labor values can also miss one of Marx's key aims: theorizing the central tendencies of capitalist economies. This is one of the descriptive strengths of Marx's theory of value.

A. Marx's Long-Period Method

The formation of central tendencies in capitalist economies and their relationship to Marx's theory of value can be brought into clear focus through Marx's 'long-period method' (LPM) (Foley 2011*b*; Garegnani 1970, 1976, 1984; Kurz and Salvadori 1995). The LPM is an approach to understanding the inner laws and tendencies of capitalist economies that is shared across the classical political economists notably Smith (2000) and Ricardo (1951)—and Marx. The LPM makes the abstract assumptions of labor and capital mobility to set aside temporary frictions in markets and/or historical abnormalities that may exist in reality so that capitalism's pure, or ideal, form can be analyzed—"the form that corresponds" to its "concept" (Marx 1981, p.291). The abstract assumptions of the LPM include consideration of a sufficient length of time so that the "real inner laws of capitalist production" can be seen as the "outcome of a whole series of protracted oscillations, which require a good deal of time before they are consolidated and balanced out" (Marx 1981, pp.266, 291). The sufficient length of time—i.e. the 'long-period'—is important for abstracting from vestiges of earlier modes of production that may affect capitalist social relations and labor and capital mobility.

In capitalist economies composed of many industries, workers expend labor effort producing value and surplus value, and are mobile across industries. Capitalists make decisions of how to allocate capital (i.e. where to invest capital), and capital is similarly mobile across industries. Labor mobility across industries induces a tendency for rates of exploitation to equalize as workers migrate across industries in response to variations in the amount of effort they need to expend relative to their compensation (Marx 1981; Foley 2016, 2018; Cogliano 2023a). If labor is assumed to be homogeneous and the length of the working day is uniform, labor mobility manifests as equalization of the wage rate, w, which is typical in many treatments of Marxian value theory, as in section II.¹² The movement of workers across industries is viewed as an open-ended process and the equalized rate of exploitation emerges as a center of gravity for ongoing fluctuations in rates of exploitation. Capital mobility means that capital moves from one industry to another in search of higher profit rates (or returns on investment) and so profit rates tend to equalize across industries. Capital exits industries with low profit rates and enters those with higher profit rates. The constant movement of capital across industries causes profit rates to continuously fluctuate and an equilibrium profit rate emerges as a center of gravity for these fluctuations (Marx 1981, pp.297-298). The equalization of the profit rate captures the classical-Marxian competitive process and leads to the formation of prices of production.

Marx elevates the tendencies for rates of exploitation and profit rates to, independently, equalize to the level of "economic laws" (Marx 1981, pp.275, 291). However, the long-period equilibrium with uniform rates of exploitation and profit rates is not necessarily a realized, or attainable, state of the

 12 See Cogliano (2023*a*) for detailed discussion of Marx's (1981, pp.241-242, 275) ideas about labor mobility and equalization of rates of exploitation, including the degree to which these ideas appear in the dual-system framework.

world given the concrete realities of capitalist economies. In reality, economies are in a constant state of flux and could operate above or below this equilibrium for extended periods of time. Similarly, historical and institutional barriers to labor and capital mobility could exist at any given time and take a variety of forms. In light of possible disruptions to labor and capital mobility, the LPM provides an analytical reference point for understanding the dynamics of capitalist economies—i.e. it captures the "law-like form" of capitalist economies (Marx 1981, p.291).

The abstract approach of the LPM brings clear focus to the inner laws and tendencies of capitalist economies so they can be understood independent of superficial phenomena. Much of the explanatory power of the Marxian perspective in describing underlying processes of capitalism—at a theoretical level—stems from the LPM and its abstract assumptions. The LPM also helps to highlight human aspects of Marx's ideas and the central role of human activity in describing the behavior of capitalist economies.

B. Circulation and Realization: Circuits of Capital and Revenue

One process that is helpfully framed by the LPM is the circulation and realization of value. The rejection of a concept of labor values by skeptics misses the importance of the circulation and realization of value, as well as competition, in Marx's theory (Shaikh 1981, p.300). In Marx's (1981, Ch.9) famous tables in volume three of *Capital*—which motivate the transformation problem debates—he shows how surplus value produced through the expenditure of productive labor effort across industries is realized as profit through the processes of circulation and competition. Capitalist competition tendentially leads to equalization of profit rates, and this equalized profit rate means that surplus value is realized in proportion to capital advanced. However, as Marx shows, surplus value is produced in proportion to the living labor engaged in an industry. Thus, as emphasized by Shaikh (1981, 1982) and Foley (2013), there are can be important differences between value and surplus value created at the point of production and their realized form of prices and profit.¹³ Discrepancies between labor values and prices capture transfers of value taking place between the production and realization of value.

Shaikh (1981, 1982, 1992) provides a detailed discussion of how value transfers across the circuits of capital and revenue can lead to realized profit being different from actual surplus value produced. Total value produced can be conserved, even when prices are not proportional to labor values, and the particulars of the circulation and realization of commodities can cause surplus value to be different from profit. Shaikh (1981, 1982) illustrates this through an example of an economy with three departments: means of production, worker consumption goods, and capitalist consumption goods. In the first two departments transfers of value due to price-value deviations (i.e. $\mathbf{p} \neq \phi \mathbf{v}$) remain contained, or are accounted for, within the circuit of capital. However, price-value deviations in the third department, for example if price is below (above) value, cause a decline (increase) in value within the circuit of capital and an increase (decrease) in the value in the circuit of revenue, i.e. value is transferred between the circuits. This transfer causes total surplus value to differ from total realized profit, and the extent of these transfers depends on how much surplus value capitalists consume.

In Shaikh's approach, this "transfer of value between the circuit of capital and the circuit of revenue, through the process of exchange... explains why price-value deviations can give rise to deviations between the sum of profits and the sum of surplus-values, *without violating the law of the conservation of value through exchange* [original emphasis]" (Shaikh 1981, p.285). An implication of Shaikh's example is that the proportionality between total surplus value and total profit need not necessarily hold.¹⁴ Thus, the rejection of labor values and labor accounts are possibly misguided and overlooks important descriptive aspects of the LTV.

A key point in Shaikh's presentation is that money expressions of value may necessarily differ from their underlying labor values, and, relatedly, dual-system calculations of prices from a given technology and wage obscure important processes of circulation and realization, and the role of labor values in these. Stated slightly differently, Steedman's (1977) argument that labor values are redundant because

 $^{^{13}}$ See Cogliano (2018, 2023*a*) for recent discussions of the importance of tracking differences between surplus value production and its realization.

 $^{^{14}}$ It is possible to argue that capitalist consumption of surplus value should be classified as part of total profit. However, debates along these lines are beyond the scope of this paper.

99

prices and labor values can be calculated from the same given data overlooks human aspects of Marx's ideas, particularly the production of value through the labor process and how prices of production form through processes of competition and realization.¹⁵ Marx's LPM helps to highlight these processes in its description of the formation of central tendencies, and it supports the relevance of the LTV as a descriptive theory.

C. Money, Exploitation, and Marx's Identities

Shaikh's analysis of the circuits of capital and revenue raises a crucial question: in light of the transformation problem debates and skepticism toward Marx's theory of value, what aspects of Marx's theory and Marx's identities are most relevant, particularly in maintaining the human and descriptive dimensions of the LTV? The NI provides some insight into this question.

As mentioned in section III.A, in the Marxian framework, broadly speaking, the expenditure of human labor effort is the source of value, and in capitalist economies value takes on form of—or is realized as—money. The NI emphasizes this connection between human labor effort and money value added (Duménil 1980; Foley 1982, 1986; Mohun 1994, 2004). Specifically, the NI treats the "labor theory of value as the claim that the money value of the whole mass of net production of commodities expresses the expenditure of the total social labor in a commodity-producing economy" (Foley 1982, p.37). The connection between labor effort and money value is also intuitive based on Marx's description of *labor-power* as the "capacity for labour" and the "expenditure of human brains, muscle, nerves, hands, etc.", and the measurement of labor-power in units of socially necessary labor time, which comprise the value of commodities (Marx 1976, p.129, 134, 270).

In the NI, the relationship between money value and labor effort is summarized in the monetary expression of labor time (MELT), which has the units money value added per unit of labor time. The MELT allows conversion between magnitudes measured in monetary units and those measured in labor time, or labor values, in the aggregate or average cases. Another characteristic of the NI is that it treats the conservation of aggregate value across circulation and exchange as an axiom, or basic principle of the LTV. Letting μ denote the MELT, under the conservation axiom: $\mathbf{py} = \mu \mathbf{vy} = \mu \mathbf{Lx}$. Thus, μ facilitates proportionality between total price and total value (in terms of net output), as well as total living labor engaged in the economy, \mathbf{Lx} . In the *n*-commodity example used here, μ can be defined as $\mu = \mathbf{py}/\mathbf{Lx}$.

The NI also defines the value of labor-power as $w^{\mu} = w/\mu$, whereas the value of labor power under the dual-system is the labor embodied in the subsistence bundle, vb. Under the definitions of the NI, one can then also convert between aggregate profit (measured in money units), $\Pi = \mathbf{p}(\mathbf{I} - \mathbf{A})\mathbf{x} - w\mathbf{L}\mathbf{x}$, and aggregate surplus value (measured in labor time), $S = (\mathbf{p}(\mathbf{I} - \mathbf{A})\mathbf{x} - w\mathbf{L}\mathbf{x})/\mu = \Pi/\mu$.

As outlined here, the NI allows two of Marx's identities to hold: total price is proportional to total value, and total profit is proportional to total surplus value.¹⁶ However, rather than these identities holding as the result of a theoretical apparatus of prices and labor values, the identities hold by definition whether or not prices are proportional to labor values. Thus, the NI has the advantage that it "is completely general, in that it is consistent with *any* theory of price formation (including, but not restricted to, theories of profit-rate equalizing prices of production) [original emphasis]" (Foley 2000, p.23). This is not uncontroversial in the transformation problem literature, and there are some important differences between the NI and dual-system.

One key difference is treating *net product*, \mathbf{y} , as the relevant measure of output (or aggregator) in the proportionality of total price and total value. Some have taken issue with this aspect of the NI since Marx's (1981, Ch.9) treatment of labor values and prices includes proportionality between gross output

 $^{^{15}}$ See Shaikh (2016, pp.212-213, fn.7) for an additional example that highlights the human and social aspects of Marx's ideas that can be overlooked by dual-system approaches. See also Shaikh (1977, 2016) for more on Shaikh's approach to price-value deviations.

¹⁶Equality of the value and price profit rates can hold under the NI if one uses μ to convert money constant capital to its labor time equivalent. However, this treatment of constant capital and the relevance of a value profit rate hold an unsettled status in the literature following the NI. In the case of constant capital, Foley (2000, 2018) raises questions about the meaning of the labor time equivalent of constant capital, and in the case of the value profit rate, Mohun (2004) raises doubts about the concept itself.

and total value, which is also typical in the dual-system.¹⁷ However, the focus on net product in the NI, arguably, maintains a more clear connection between value production and the expenditure of labor effort than other approaches. This connection can be seen in the aggregate relationship $\mathbf{py} = \mu \mathbf{vy} = \mu \mathbf{Lx}$. where the total value produced, **py**, is proportional to the total living labor effort expended in commodity production, Lx. Under the dual-system such a relationship does not hold in general, $\mathbf{px} \neq \mathbf{vx} \neq \mathbf{Lx}$, and the connection between value creation and labor effort is not as clear.

Under the NI's definitions, total profit and total surplus value are also proportional, and more importantly for our current discussion, surplus value corresponds to unpaid labor effort (Foley 2016). Since $\mathbf{py} = \mu \mathbf{Lx}$ and $S = (\mathbf{py} - w\mathbf{Lx})/\mu$, surplus value can also be written as $S = (\mathbf{Lx} - w\mathbf{Lx})/\mu$. The NI's definitions of surplus value and the value of labor power define exploitation "in terms of the monetary relations of the capitalist system of production" rather than simply equating the existence of exploitation with "the existence of a surplus product over the necessary consumption of a society" as in the dual-system (Foley 1982, p.43).¹⁸ The "analytical danger" of equating exploitation with a surplus product is that a surplus product is a technical necessity for a growing economy, and a surplus product alone does not describe "the capitalist form of exploitation through money and the market" (Foley 1982, p.43).

A related conclusion of the transformation problem debates and the dual-system approach is that exploitation has limited meaning beyond being a determinant of positive aggregate profit by ensuring the existence of a surplus product. The association of positive profits with the existence of exploitation is summarized in the Fundamental Marxian Theorem (FMT) (Morishima 1973). An added strength of the NI's definition of exploitation is that exploitation is not just a determinant of aggregate profit or relegated to the FMT.¹⁹ The NI's understanding of "exploitation in terms of the monetary relations of the capitalist system of production" allows exploitation to more readily be seen as "a factual description of a socio-economic relationship" (Foley 1982, p.43; Dobb 1973, p.145). Exploitation in the NI also, arguably, better frames class struggle as "a struggle not to eliminate the surplus product, but to transform the social relations that make the social surplus take the form of surplus value" (Foley 1982, p.43).

The NI's emphasis on the relationship between money value and labor effort leads to multiple conclusions that highlight the descriptive strengths of Marx's theory of value. In particular, that value creation is clearly associated with the expenditure of human labor effort, and that stating exploitation in terms of money relations captures real social relationships of capitalism. These also highlight the human dimension of the LTV and the importance of understanding capitalism through "personal participation through labour" (Sen 1978, p.177). It may be argued that the NI shifts the emphases found in the dual-system, and the approach does have its critics. However, the NI preserves key relationships in Marx's original work that align well with the descriptive interpretation of the LTV and Marx's focus on understanding key processes of capitalism.

Competition and Profit Rates IV.

The theory of competition and the behavior of profit rates is an area where the descriptive interpretation of the LTV is helpful for highlighting fundamental processes of capitalist economies. Competition in the classical-Marxian framework describes processes of the allocation of capital, the formation of equilibrium prices, and open-ended gravitation or ongoing fluctuations—as discussed in section III.A. The classical-Marxian competitive process is emphasized by Shaikh (1978, 1979, 1980b, 2008, 2016) across multiple contributions, and these interventions bring clarity to the competitive process in capitalist economies. As Shaikh (2016) and others—i.e. Clifton (1977)—explain, neoclassical (or conventional) theories of competition cannot capture the *process* of competition in the same way the classical-Marxian theory can. Even theories of market structures differing from perfect competition, such as imperfect

As long as $\mathbf{vb} < 1$, $\frac{1}{\mathbf{vb}} - 1 > 0$. ¹⁹See Veneziani and Yoshihara (2015, 2017*a*,*b*) and Yoshihara and Veneziani (2018) for detailed analyses of the relation-

 $^{^{17}}$ Details of the debates about use of net product versus gross product are beyond the current scope. See Foley (2000), Moseley (2000), and Fine, Lapavitas and Saad-Filho (2004) for discussion along these lines. ¹⁸The rate of exploitation in the dual-system can be written as $\frac{\mathbf{v}(\mathbf{I}-\mathbf{A}-\mathbf{b}\mathbf{L})\mathbf{x}}{\mathbf{v}\mathbf{b}\mathbf{L}\mathbf{x}}$, which reduces to $\frac{1}{\mathbf{v}\mathbf{b}}-1$ if $\mathbf{v} = \mathbf{L} (\mathbf{I}-\mathbf{A})^{-1}$.

ship between exploitation and profits in different approaches to Marxian value theory.

competition, can fall short since they are simply attempts to "make standard theory more realistic" and do not focus on processes of competition (Shaikh 2016, p.329).

A key contribution by Shaikh (1978, 1979, 1980*b*, 2008, 2016) is the refinement of the classical-Marxian theory of competition as the theory of "real competition".²⁰ In the theory of real competition individual capitalists compete both within and across industries, as competition is a "war of all against all" (Marx 1976, p.477). One feature of the theory of real competition is that it recognizes that individual industries are characterized by a structure of profit rates and capitalists within industries compete through costs and choices of techniques (Botwinick 1993, p.131). The price of the good produced by an industry will be governed by the "average" capital (Marx 1981, p.273): "This production price, as we explained earlier, is determined not by the individual cost price of any one industrialist producing by himself, but rather by the price that the commodity costs on average under the average conditions for capital in that whole sphere of production" (Marx 1981, p.779).

These "average conditions" that determine the price for an industry also determine the entry point for new capitalists in the industry. When capitalists compete across industries they enter industries under the "generally reproducible conditions" of production (Shaikh 2008, p.167). Any industry might be characterized by capitalists with some kind of unique cost advantage that yields them a higher-thanaverage profit rate, but the advantage possessed by these capitalists may not be replicable. For instance, Marx (1981, pp.779-785) uses the example of a capitalist with access to a waterfall as a source of power while other capitalists in the same industry rely on steam-engines for power since they do not have access to—or property rights over—the waterfall. Thus, new entrants into this industry will not be able to use the waterfall for power, they will need to use steam-engines or whatever the prevailing "average social conditions of production" might be (Marx 1981, p.780). In the theory of real competition, capitalists operating at these average conditions are defined as "regulating capitals" (Shaikh 2008, 2016; Botwinick 1993; Tsoulfidis and Tsaliki 2005).

Regulating capitals reflect the average conditions of production within and across industries, and they determine the profit rate on new investment in production. Thus, competition within industries and capital movement across industries will cause "the turbulent equalization of actual rates of profit on new investments" or "regulating profit rates" (Shaikh 2008, p.168).²¹ The theory of real competition and the notion of a regulating capital have empirical merit, see for instance Mokre and Rehm (2020) and Tsoulfidis and Tsaliki (2005), but for the current focus, they also fit in with the descriptive approach to Marx's theory. Specifically, the theory of how competition *actually* operates within and across industries and what is a *realistic* profit rate to consider provide a granular description of the competitive process that other theories lack.

Competition and turbulent equalization of profit rates are fundamental processes in capitalist economies, and the classical-Marxian approach has a great deal of explanatory power for understanding these processes. The explanatory power of the classical-Marxian approach is not limited to description. Recent empirical treatments by Scharfenaker and Semieniuk (2017), Scharfenaker and Foley (2017), and dos Santos and Yang (2020), for example, show that distributions profit rates for firms across industries are "sharply peaked". These contributions focus on measurements of profit rates that are more standard in the classical-Marxian literature—i.e. returns on total invested capital as opposed to returns on new investment, as in studies of regulating capitals—and they develop novel empirical methods using tools of statistical mechanics.

A theme in the results is that profit rates fit sharply-peaked distributions with an understandable degree of dispersion around the peak, which can be seen as a reflection of the competitive process. Thus, statistically speaking, there is solid evidence for profit rate equalization with the equilibrium serving as a center of gravity. The information-theoretic approach deployed by the aforementioned contributions also has descriptive elements that align well with the overall focus of this paper. These approaches are able to develop a robust theory of how capital mobility decisions are made, which are the decisions underlying the actual process of competition in capitalist economies, and provide a description of how central tendencies like profit rate equalization are attained. Standard theories of competition do not

²⁰See also Botwinick (1993) and Tsoulfidis and Tsaliki (2005).

 $^{^{21}}$ Regulating profit rates are sometimes also referred to as "incremental profit rates" or "incremental rates of return" (Duménil and Lévy 2012; Mokre and Rehm 2020).

capture underlying processes of capital mobility and competition, and the typical critiques of Marx's theory of value miss its descriptive validity.

The competitive process also plays a role in the issues of labor values and prices shown in the context of the transformation problem debates in sections II and III. Recall that Marx's (1981, Chs. 9, 10) identities are derived from a series of tables in volume three of *Capital* where he shows how surplus value is conserved in the competitive process and the formation of a general rate of profit and prices of production. Surplus value is produced in proportion to the living labor engaged in an industry, and surplus value produced in relatively labor-intensive industries is 'transferred' to relatively capital-intensive industries via competition such that a uniform r holds and prices of production, \mathbf{p} , form. Thus, the competitive process *creates* the differences between labor values, \mathbf{v} , and prices that lie at the heart of the transformation problem.

Marx (1981) is careful to point out how the process of competition has a distorting influence:

In competition, therefore, everything appears upside down. The finished configuration of economic relations, as these are visible on the surface, in their actual existence, and therefore also in the notions with which the bearers and agents of these relations seek to gain an understanding of them, is very different from the configuration of their inner core, which is essential but concealed, and the concept corresponding to it [original emphasis] (Marx 1981, p.311).

The process by which prices of production form and the profit rate equalizes have this distorting effect and seem to "contradict both the determination of value by labour-time and the nature of surplus-value as consisting of unpaid surplus labour" (Marx 1981, p.311). This is precisely the issue on which critics of Marx's theory of value get stuck. The seeming lack of connection between **p** and **v** is not a problem with Marx's theory, it is a feature consistent with his vision of the competitive process. In fact, at a theoretical level, labor values and prices must be different. This difference is necessary not only to account for competition and profit rate equalization, it is also necessary for Marx's theory to be able to explain the more general case of unproductive activities (or unproductive labor), where the entirety of conventionally-measured value added for some industries (or industry-level GDP) is surplus value produced by other industries (Cogliano 2018, 2023*a*,*b*). The case of unproductive activities is discussed further in section V.

Despite the inconsistency between \mathbf{p} and \mathbf{v} at a theoretical level, measures of labor values and prices *can* be close empirically. As Shaikh (1984, 1998, 2016), Ochoa (1989), Chilcote (2004), Tsoulfidis and Maniatis (2002), and Işikara and Mokre (2022) show, *direct prices*—which are prices proportional to estimates of labor values from input-output data—are strongly correlated with observed market prices. These results have a great deal of predictive value in that the relationship between direct and market prices implies that the underlying labor content of commodities bears relevance for prices. These results are also a forceful counter to those who might criticize labor values as being 'metaphysical', or lacking in substance for applied analyses. The possible tension between the theoretical inconsistency of labor values and prices and the empirical studies mentioned immediately above is beyond the scope of this paper.²² The empirical support for labor values provides a vindication for Marx's ideas, but labor values are also important for other aspects of Marxian analyses, as is discussed in section VI.

V. Labor Mobility and Social Labor

The obfuscating role of competition poses interesting problems for understanding the sources of value and surplus value production. As discussed in section III.C, the NI maintains a clear connection between total living labor and aggregate value production. In the aggregate, a relationship between human labor effort and value creation seems straightforward. However, at disaggregated levels this becomes more complicated. As Marx (1981, p.311) notes—and as discussed in section IV—the competitive process seems to undermine labor effort as the source of value and surplus value, and the industry level surplus

²²See Torres-González and Yang (2019) and Torres-González (2022) for recent studies of the statistical properties of input-output data for the United States, which speak to some of the underlying reasons why direct prices and market prices can be empirically close. These studies find some tendency toward uniformity of organic compositions of capital, \mathbf{A}_j/l_j for $j = 1, \ldots, n$, does exist and this can serve as part of an explanation for for the closeness of labor values and prices.

value produced can be different from surplus value realized as profit. Surplus value is produced in proportion to the living labor engaged in an industry, but in the long-period equilibrium, is realized in proportion to the capital advanced such that a uniform r holds.²³ The competitive process thus poses a problem for directly observing surplus value production at the industry level, or at more disaggregated levels.

In capitalist economies the money form of prices and profits are what can be directly observed, or are "visible surface phenomena", and "surplus-value and the rate of surplus-value are, relative to this, the invisible essence to be investigated" (Marx 1981, p.134). The problem of analyzing surplus value production at the industry level, as well as a broader understanding of the sources of value creation, is more of an *'inverse transformation problem'* where one needs to move from observable prices and profits (measured in money) to the underlying labor accounts of labor values and surplus value (measured in units of labor time). The inverse transformation problem was first identified by Robinson (1950), with the term first being used by Morishima and Seton (1961)—although the broader notion of uncovering labor accounts from observable data was implied even earlier by Rubin (1990). More recently, Foley (2011*a*) and Cogliano (2018, 2023*a*,*b*) have discussed the inverse transformation in detail, and proposed methods for undertaking the procedure.

An approach to the inverse transformation follows from the LPM and the abstract assumption of labor mobility. The labor and capital mobility in the LPM, respectively, lead to the independent tendencies for equalization of rates of exploitation and profit rates (Foley 2011 b; Cogliano 2023 a). Equalization of rates of exploitation means that workers evaluate the effort they need to expend in any job and the wages they receive in exchange for their effort. As Foley (2016) shows, balancing the ratio of labor effort to income is consistent with an equalized rate of exploitation, since surplus value represents surplus labor effort. Furthermore, the connection between ratios of labor effort to wages and rates of exploitation is intuitive considering Marx's (1981, pp.241-242) source for his ideas about labor mobility in Smith's (2000, Ch. 10) description of the equalization of the "advantages and disadvantages of labor". The connections between Smith and Marx, as well as equalization of rates of exploitation and the connection to labor effort are discussed in detail by Cogliano (2023 a).

Marx's equalized rate of exploitation—abbreviated as EQRE—in conjunction with the conservation of value of the NI allows for an inverse transformation to be performed to recover surplus value production at the industry level. The aggregate rate of exploitation can be observed and calculated as the rate of surplus value, $\varepsilon = \frac{\mathbf{p}(\mathbf{I}-\mathbf{A})\mathbf{x}-w\mathbf{L}\mathbf{x}}{w\mathbf{L}\mathbf{x}} = \frac{\Pi}{w\mathbf{L}\mathbf{x}}$. Under the definitions of the NI, the rate of exploitation is known from ε by converting money magnitudes to their labor time equivalents using μ . The rate of exploitation is $e = \frac{(\mathbf{p}(\mathbf{I}-\mathbf{A})\mathbf{x}-w\mathbf{L}\mathbf{x})/\mu}{w\mathbf{L}\mathbf{x}/\mu} = \frac{\Pi/\mu}{w\mathbf{L}\mathbf{x}/\mu} = \frac{S}{w\mathbf{L}\mathbf{x}/\mu}$, and $e \equiv \varepsilon$. Marx also assumes that e is uniform across productive industries, or $e_j \equiv e$, all j, as a result of labor mobility. Equivalency of e and ε is shown in Marx's original tables in volume three of *Capital*, and as Cogliano (2023*a*) details, can arguably be seen as an additional 'identity' that much of the existing literature misses. Similarly, the rationale for uniformity of e_j is largely unrecognized in existing literature, even though it can be viewed as a "hidden part of the *definition* of 'value,' [original emphasis]" (Morishima and Seton 1961, p.206).²⁴

If $\mu = \mathbf{py}/\mathbf{Lx}$, the rate of exploitation can be written as $e = \frac{\mu}{w} - 1$, and solving for the wage rate, $w = \mu\left(\frac{1}{1+e}\right)$. Let **s** be the industry-level surplus value produced per unit of output. Under the NI, surplus value production at the industry level can be written as the difference between labor performed and the labor time equivalent of the industry wage bill: $\mathbf{s} = \mathbf{L} - \frac{w}{\mu}\mathbf{L}$. Since $w = \mu\left(\frac{1}{1+e}\right)$, $\mathbf{s} = \mathbf{L}\left(1 - \frac{1}{1+e}\right)$, and again, surplus value production is proportional to the living labor engaged in

²³This is apparent in both the dual-system framework shown in section II, and in the NI. In the dual-system the surplus value produced across industries per unit of output is $\mathbf{v} (\mathbf{I} - \mathbf{A} - \mathbf{b} \mathbf{L})$, and with $\mathbf{v} = \mathbf{L} (\mathbf{I} - \mathbf{A})^{-1}$, this becomes $(1 - \mathbf{v} \mathbf{b}) \mathbf{L}$, whereas the profit (or realized surplus value) is $\mathbf{p} (\mathbf{I} - \mathbf{A}) - w \mathbf{L}$.

²⁴See Cogliano (2023*a*) for an in-depth discussion of both aspects of Marx's EQRE, including its intellectual roots in the work of Smith (2000), and how it applies in both the NI and dual-system. In the dual-system, $\varepsilon \neq e$ since $\mathbf{p} \neq \phi \mathbf{v}$; formally, $\frac{\mathbf{p}(\mathbf{I}-\mathbf{A})\mathbf{x}-w\mathbf{L}\mathbf{x}}{w\mathbf{L}\mathbf{x}} \neq \frac{\mathbf{v}(\mathbf{I}-\mathbf{A}-\mathbf{b}\mathbf{L})\mathbf{x}}{\mathbf{v}\mathbf{b}\mathbf{L}\mathbf{x}}$. However, uniformity of e_j across productive industries is built into standard definitions of labor values, with industry-level rates of exploitation across the *n* productive industries all being $\frac{1}{\mathbf{v}\mathbf{b}} - 1$. See Cogliano (2023*a*) for further elaboration.

an industry. This reflects a foundational precept of the NI: the abstract labor content of production is proportional to the wage bill, or its labor time equivalent.

Recognizing the theoretical importance of labor mobility and role of e in estimating surplus value production is especially important in the more general case of unproductive industries. Contributions by Basu and Foley (2013), Foley (2011c, 2013), Mohun (2006, 2009, 2013, 2014), Moseley (1988, 1990, 1991), Paitaridis and Tsoulfidis (2012), and Shaikh and Tonak (1994), for example, show how considering a distinction between productive and unproductive industries—as well as possibly applying a similar distinction within industries—changes how categories like surplus value need to be calculated. Basu and Foley (2013) and Foley (2011c, 2013) provide detailed explanations of how entire industries—particularly those like the financial services, insurance, and real estate industries, among others—might be considered unproductive and "value-realising" as opposed to "value-adding".

If entire industries are unproductive they do not produce surplus value or any value at all, and the whole of their conventionally-measured value added (or industry-level GDP) is surplus value produced elsewhere in the economy. As Foley (2013, p.259) explains, "because competitively-determined prices of commodities may deviate from proportionality to the commodities' productive labor content, surplus value is not necessarily realized in the productive enterprises in which it is generated". Unproductive industries still participate in the competitive process that tends to equalize profit rates, thus their activities will contribute to the obfuscating effects of competition and complicate a clear understanding of labor effort as the source of value.

The NI and Marx's EQRE are further helpful for analyzing surplus value production in this more general case of unproductive industries. Suppose an economy has m total industries numbered $k = 1, \ldots, m$, with n < m productive industries denoted by $j = 1, \ldots, n$ and m - n unproductive industries denoted by $h = n + 1, \ldots, m$. Calculating ε to find e and perform an inverse transformation procedure now requires treating the conventionally-measured value added of the m - n unproductive industries as part of the total surplus value, $S = \Pi/\mu$, but once this is done, ε can be found as the ratio of total profit to the total wages in productive industries (Cogliano 2018, 2023b). This calculation of ε will still be equivalent to e and uniformity of e_j across productive industries can be assumed to find industry-level surplus value production. Under the NI, the total abstract labor content of an industry's production is proportional to the wage bill, thus the wage bill reflects the total abstract living labor engaged in an industry. With this in mind, and because surplus value production is proportional to the living labor engaged in an industry, industry-level surplus value production can be calculated as $e \frac{W_j}{\mu}$, where W_j denotes the wage bill of productive industry j.²⁵ Using the matrix and vector notation from earlier, industry-level surplus value production in this case could also be written as $ew \mathbf{L}\hat{\mathbf{x}}$, where $\hat{\mathbf{x}}$ is the $n \times n$ diagonal matrix of the vector \mathbf{x} .

Marx's EQRE is not the only assumption that can be made to perform an inverse transformation procedure. There are also open questions as to the appropriate time horizon over which to assume this equalization might take place. However, *some* assumption needs to be made about the distribution of rates of exploitation to estimate labor accounts like surplus value and labor values from available data gathered in terms of money prices (Duménil, Foley and Lévy 2009; Foley 2011*a*; Cogliano 2018, 2023*a*,*b*). Szumski (1989) and Rieu and Park (2020, 2022), for example, explore alternatives to Marx's EQRE in undertaking an inverse transformation.²⁶

It is also worth noting that the inverse transformation procedure—as outlined here—as well as the conversions between money and labor time magnitudes in the NI do not require that prices be equilibrium prices or prices of production: they can be performed with any given prices. Thus, the NI and EQRE allow estimation of surplus value production in very general cases.

The NI and Marx's EQRE provide a way to uncover surplus value production and labor effort as the source of value in general. The approach outlined here builds on Foley's (1982; 1986; 2000; 2005; 2011 b) contributions to understanding the role of money in Marx's analysis, and offers a new role for labor accounts in Marxian value theory. Because surplus value production at the industry level is proportional

 $^{^{25}}$ See Cogliano (2018, 2023*b*) for recent applications of this method for estimating surplus value production to the case of the United States.

 $^{^{26}}$ The assumption of Marx's EQRE is at least implicit in the empirical treatments of labor values discussed in section IV.

to the living labor engaged in an industry, estimating surplus value using an inverse transformation method speaks to the allocation of productive social labor in capitalist economies. This perspective also reframes Marx's theory of value as a theory of the allocation of social labor, as opposed to a theory of prices (Foley 2016, 2018; Cogliano 2023a). This fits with the overall descriptive approach put forward in section I, where Marx's ideas about labor mobility and equalization of rates of exploitation can be seen as providing a description of the processes of the allocation of social labor that other theories do not capture.

This perspective further helps to clarify a link between labor and capital mobility and the formation of long-period prices, and connects to ideas found in Rubin (1990), particularly those about the determination of value and social need—or demand.²⁷ Seeing the LTV as a theory of the allocation of social labor shows how social need and aggregate demand can be important in Marx's theory of value, despite the common view that aggregate demand plays no role in Marx's work.

Rubin (1990, Ch.17) provides an overview of how the "volume of demand" plays a role in determining the amount of labor that is socially necessary to produce various commodities: "The multiplication of the value per unit of product (which is determined by the technical conditions of production) times the number of units which will be sold at the given value, expresses the social need which is able to pay for the given product [emphasis removed]" (Rubin 1990, pp.195-196). The social need for any particular commodity factors into the overall amount of social labor dedicated to production of that commodity. Extending this to all of capitalist production—or the economy as a whole—important connections between framing Marx's theory of value as a theory of the allocation of social labor and aggregate demand become more apparent: overall demand plays an important role in regulating quantities of "labor actually expended in given [branches] of industry" (Rubin 1990, p.185).

Rubin's approach is consistent with what is outlined above and the descriptive interpretation of the LTV. Rubin (1990, p.223) sees Marx's theory of value as describing capitalist economies as systems of "distributed labor" and "distributed capitals" in "dynamic equilibrium", where a key aim is to "see under the visible process of distribution of *capital* the invisible process of the distribution of social *labor* [original emphases]". Being able to connect an understanding of the distribution of social labor to demand is crucial for developing robust accounts of the dynamics of capitalist economies, particularly the organization of production and economic crises. The descriptive interpretation of the LTV, as well as ideas from the NI and Rubin, provide a promising framework for such analyses.

VI. Labor Accounts and Accumulation

There are other recent contributions that—similar to those in section V—reconsider the role of labor accounts in Marxian value theory. Contributions by Flaschel, Franke and Veneziani (2013), Cogliano et al. (2018), and Cogliano, Veneziani and Yoshihara (2022), for example, argue that labor values can serve as indicators of industry-level labor productivity, which are more reliable than conventional measures. The overarching aim of these contributions is similar to that of Cogliano's (2023 a) and Foley's (2016) perspective that understanding labor mobility and surplus value production speaks to the allocation of social labor: industry-level labor productivity reflects the allocation of social labor and value-creating capacity across capitalist economies. These contributions also align with a descriptive approach to the LTV.

The arguments made by Flaschel, Franke and Veneziani (2013), Cogliano et al. (2018), and Cogliano, Veneziani and Yoshihara (2022) are based on Flaschel's (1983) concept of "actual labor values", which are defined as the average total labor content of commodities. This notion of labor values is conceptually similar to the NI, close to Shaikh's (1984; 1998; 2016), and notably different from definitions of labor values used by others, for example, Morishima (1974), Morishima and Catephores (1978), and Steedman (1977).²⁸

 $^{^{27}}$ Thanks are owed to an anonymous referee for highlighting the relationship between Rubin's approach and that outlined earlier in this section.

 $^{^{28}}$ Morishima (1974) and Morishima and Catephores (1978) define labor values as the minimum amount (or efficient amount) of labor needed to produce a commodity. Steedman (1977) defines labor values as standard Leontief employment multipliers. See Cogliano et al. (2018) for more detailed discussion of different concepts and definitions of labor values, and see Yoshihara and Veneziani (2023) for an axiomatic treatment of measurements of labor content.

As technical progress takes place, driven by competitive pressures to accumulate, the labor content of commodities necessarily changes over time. Technical changes that are labor-saving and increase profitability will tend to reduce the overall labor content—or labor values—of commodities. Thus, over time, it should be expected that labor values exhibit a downward trend. Flaschel, Franke and Veneziani (2013) refer to this as the "Law of Decreasing Labor Content" (LDLC), and show that it holds for estimates of labor values based on German input-output data, while Cogliano, Veneziani and Yoshihara (2022) confirm the LDLC holds using U.S. input-output tables.

The changes in labor values captured by the LDLC act as good indicators of changes in labor productivity, as Marx (1976, p.131) notes "the greater the productivity of labour, the less the labourtime required to produce an article". Suppose that for some industry j an initial production technique can be described by (\mathbf{A}_j, l_j) and the labor value of a unit of output of industry j is $v_j = \mathbf{v}\mathbf{A}_j + l_j$. Also suppose that a new more profitable technique is denoted by (\mathbf{A}'_j, l'_j) with corresponding labor value $v'_j = \mathbf{v}'\mathbf{A}'_j + l'_j$, such that $v'_j < v_j$. This decrease in the labor value of output means that a unit of labor time applied in industry j has become more productive with the change from (\mathbf{A}_j, l_j) to (\mathbf{A}'_j, l'_j) . This Marxian notion of labor productivity can be represented by the inverse of labor values, i.e. $\rho_j = 1/v_j$ and $\rho'_j = 1/v'_j$ with $\rho'_j > \rho_j$.

and $\rho'_j = 1/v'_j$ with $\rho'_j > \rho_j$. Labor values, or $\rho = 1/\mathbf{v}$, are more reliable indicators of industry-level labor productivity than labor productivity measures based on conventionally-measured value added. Following Marx's arguments about the distorting effects of competition summarized in section IV and discussed in section V, conventional value added at the industry level (or GDP at the industry level) will include realized surplus value in the form of profit, not the surplus value that an industry actually produces. Thus, standard measures of industry-level labor productivity—i.e. conventionally-measured value added per worker—will not *really* capture labor productivity due to the differences between surplus value production and its realization in the form of profit. Marxian and conventional industry-level labor productivity can be quite different, as Flaschel, Franke and Veneziani (2013) and Cogliano, Veneziani and Yoshihara (2022) respectively show for Germany and the U.S.

Estimates of labor productivity using labor values are inherently counterfactuals given the assumptions needed in working with available input-output data—as are the estimates made using the inverse transformation method outlined in section V. However, despite their counterfactual nature, these estimates of labor accounts provide an important description of the underlying dynamics of accumulation and the allocation of social labor. Treating labor accounts in this way also, arguably, maintains important human dimensions of Marx's ideas through the emphasis on labor and value production.

VII. Conclusions and the Road Ahead

The debates around the transformation problem in the 1970s sowed doubts about the usefulness of labor accounts and Marxian value theory in general, among both mainstream and heterodox economists. These doubts place a heavy burden on those working in the Marxian tradition, as they are compelled to argue the relevance of Marx's ideas on two fronts. This is unfortunate given the merit the LTV has as a descriptive theory. In particular, the usefulness of labor accounts in explaining deep-seated processes and underlying dynamics of capitalist economies, and the human dimensions of Marx's work maintained by the descriptive approach. The works of Duncan Foley and Anwar Shaikh on Marx's theory of value make fundamental contributions along these lines.

In light of the developments in Marxian value theory in recent decades, adhering to the standard criticisms of the LTV is, arguably, no longer a tenable position. Foley's and Shaikh's contributions have done a great deal to shed light on the relevance of the LTV and labor accounts for understanding competition and profit rate dynamics, the role of money in capitalist social relations, and human labor effort as the source of value, among other topics. These contributions also leave multiple open avenues for future research in Marxian value theory. Possible areas for future work include, but are not limited to, further studies of price-value deviations and detailed studies of competitive dynamics from a Marxian perspective, the relationship between money and labor accounts, methods for estimating labor accounts from monetary data, and treating the LTV as a theory of the allocation of social labor. The contributions discussed in this paper and the promising lines for future research are a solid foundation from which to, hopefully, cast off the stale critiques of Marxian value theory.

REFERENCES

- Basu, Deepankar, and Duncan K. Foley. 2013. "Dynamics of Output and Employment in the U.S. Economy." *Cambridge Journal of Economics*, 37(5): 1077–1106.
- Botwinick, Howard. 1993. Persistent Inequalities. Princeton, New Jersey: Princeton University Press. Chilcote, Edward B. 2004. "Calculating Labour Values Empirically." In The New Value Controversy
- and the Foundations of Economics., ed. A. Freeman, A. Kliman and J. Wells. Cheltenham, UK: Edward Elgar.
- Clifton, James A. 1977. "Competition and the evolution of the capitalist mode of production." Cambridge Journal of Economics, 1(2): 137–151.
- Cockshott, W. Paul, and Allin F. Cottrell. 1997. "Labour time versus alternative value bases: a research note." *Cambridge Journal of Economics*, 21(4): 545–549.
- Cogliano, Jonathan F. 2018. "Surplus Value Production and Realization in Marxian Theory Applications to the U.S., 1990-2015." *Review of Political Economy*, 30(4): 505–533.
- Cogliano, Jonathan F. 2023a. "Marx's equalised rate of exploitation." Cambridge Journal of Economics, 47(1): 133–169.
- Cogliano, Jonathan F. 2023b. "Surplus Value and the Allocation of Social Labor." Marxism21, 20(1): 12–66.
- Cogliano, Jonathan F., Peter Flaschel, Reiner Franke, Nils Fröhlich, and Roberto Veneziani. 2018. Value, Competition, and Exploitation: Marx's Legacy Revisited. Cheltenham: Edward Elgar.
- Cogliano, Jonathan F., Roberto Veneziani, and Naoki Yoshihara. 2016. "The Dynamics of Exploitation and Class in Accumulation Economies." *Metroeconomica*, 67(2): 242–290.
- Cogliano, Jonathan F., Roberto Veneziani, and Naoki Yoshihara. 2019. "Exploitation, skills, and inequality." *Review of Social Economy*, 77(2): 208–249.
- Cogliano, Jonathan F., Roberto Veneziani, and Naoki Yoshihara. 2021. "The Dynamics of International Exploitation." University of Massachusetts Boston Working Paper 2021-02.
- Cogliano, Jonathan F., Roberto Veneziani, and Naoki Yoshihara. 2022. "Putting labour values to work." Acta Oeconomica, 72(S1): 51–79.
- Cohen, G.A. 1979. "The Labor Theory of Value and the Concept of Exploitation." *Philosophy & Public Affairs*, 8(4): 338–360.
- Desai, Meghnad. 1988. "The Transformation Problem." Journal of Economic Surveys, 2(4): 295–333.
- Dobb, Maurice. 1937. Political Economy and Capitalism. London: Routledge.
- **Dobb, Maurice.** 1973. Theories of value and distribution since Adam Smith. London: Cambridge University Press.
- dos Santos, Paulo, and Jangho Yang. 2020. "Capital Mobility, Quasi-Rents, and the Competitive Self-Organization of Distrbutions of Profitability." Advances in Complex Systems, 23(2).
- Duménil, Gérard. 1980. De la valeur aux prix de production. Paris: Economica.
- **Duménil, Gérard.** 1984. "The So-Called 'Transformation Problem' Revisited: A Brief Comment." Journal of Economic Theory, 33(2): 340–348.
- **Duménil, Gérard, and Dominique Lévy.** 2012. "Is the IROR a Plausible Approximation of the Profit Rate on Regulating Capital?" EconomicX, PSE, Paris.
- Duménil, Gérard, Duncan K. Foley, and Dominique Lévy. 2009. "A Note on the Formal Treatment of Exploitation in a Model with Heterogeneous Labor." *Metroeconomica*, 60(3): 560–567.
- Fine, Ben, Costas Lapavitsas, and Alfredo Saad-Filho. 2004. "Transforming the Transformation Problem: Why the 'New Interpretation' is a Wrong Turning." *Review of Radical Political Economics*, 36(1): 3–19.
- Flaschel, Peter. 1983. "Actual Labor Values in a General Model of Production." *Econometrica*, 51(2): 435–454.
- Flaschel, Peter, Nils Frölich, and Roberto Veneziani. 2013. "The sources of aggregate profitability: Marx's theory of surplus value revisited." Intervention – European Journal of Economics and Economic Policies, 10.
- Flaschel, Peter, Reiner Franke, and Roberto Veneziani. 2013. "Labor productivity and the Law of Decreasing Labor Content." *Cambridge Journal of Economics*, 37(2): 379–402.

- Foley, Duncan K. 1982. "The Value of Money, the Value of Labor Power and the Marxian Transformation Problem." *Review of Radical Political Economics*, 14(2): 37–47.
- Foley, Duncan K. 1986. Understanding Capital. Cambridge, MA: Harvard University Press.
- Foley, Duncan K. 2000. "Recent Developments in the Labor Theory of Value." *Review of Radical Political Economics*, 32(1): 1–39.
- Foley, Duncan K. 2005. "Marx's Theory of Money in Historical Perspective." In Marx's Theory of Money., ed. F. Moseley. New York, NY: Palgrave Macmillan.
- Foley, Duncan K. 2011a. "I. I. Rubin and the Inverse Transformation Problem." Paper presented at the Russian Academy of Sciences, Institute of Economics: Round Table on I. I. Rubin, December 15, 2011, Moscow.
- Foley, Duncan K. 2011b. "The Long-Period Method and Marx's Theory of Value." In *The Evolution* of *Economic Theory: Essays in Honour of Bertram Schefold.*, ed. V. Caspari. Abingdon, Oxon: Routledge.
- Foley, Duncan K. 2011c. "The Political Economy of U.S. Output and Employment 2001-2010." Schwartz Center for Economic Policy Analysis (SCEPA) Working Paper 2011-5.
- Foley, Duncan K. 2013. "Rethinking Financial Capitalism and the 'Information' Economy." *Review* of *Radical Political Economics*, 45(3): 257–268.
- Foley, Duncan K. 2016. "What is the labor theory of value and what is it good for?" In *Economic Theory and its History: Essays in honour of Neri Salvadori.*, ed. G. Freni, H. D. Kurz, A. M. Lavezzi and R. Signorino. Abingdon, Oxon: Routledge.
- Foley, Duncan K. 2018. "The 'New Interpretation' After 35 Years." Review of Radical Political Economics, 50(3): 559–568.
- Fröhlich, Nils. 2013. "Labour values, prices of production and the missing equalisation tendency of profit rates: evidence from the German economy." *Cambridge Journal of Economics*, 37(5): 1107–1126.
- Garegnani, Pierangelo. 1970. "Heterogeneous Capital, the Production Function and the Theory of Distribution." *Review of Economic Studies*, 37(3): 407–436.
- Garegnani, Pierangelo. 1976. "On a change in the notion of equilibrium in recent work on value and distribution: a comment on Samuelson." In *Essays in Modern Capital Theory.*, ed. M. Brown, K. Sato and P. Zarembka. Amsterdam: North-Holland.
- Garegnani, Pierangelo. 1984. "Value and Distribution in the Classical Economists and Marx." Oxford Economic Papers, 36(2): 291–325.
- Işikara, Güney, and Patrick Mokre. 2022. "Price-Value Deviations and the Labour Theory of Value: Evidence from 42 Countries, 2000-2017." *Review of Political Economy*, 34(1): 165–180.
- Kurz, Heinz D., and Neri Salvadori. 1995. Theory of Production: A long-period analysis. Cambridge, UK: Cambridge University Press.
- Marx, Karl. 1976. Capital: Volume I. New York, NY: Penguin Group.
- Marx, Karl. 1981. Capital: Volume III. New York, NY: Penguin Group.
- Mohun, Simon. 1994. "A re(in)statement of the labor theory of value." Cambridge Journal of Economics, 18(4): 391–412.
- Mohun, Simon. 1996. "Productive and Unproductive Labor in the Labor Theory of Value." *Review of Radical Political Economics*, 28(4): 30–54.
- Mohun, Simon. 2004. "The Labour Theory of Value as Foundation for Empirical Investigations." Metroeconomica, 55(1): 65–95.
- Mohun, Simon. 2005. "On measuring the wealth of nations: the US economy, 1964-2001." Cambridge Journal of Economics, 29(5): 799–815.
- Mohun, Simon. 2006. "Distributive shares in the US economy, 1964-2001." Cambridge Journal of Economics, 30(3): 347–370.
- Mohun, Simon. 2009. "Aggregate capital productivity in the US economy, 1964-2001." Cambridge Journal of Economics, 33(5): 1023–1046.
- Mohun, Simon. 2013. "Rate of profit and crisis in the US economy: a class perspective." In *Social Fairness and Economics: Economics Essays in the Spirit of Duncan Foley.*, ed. L. Taylor, A. Rezai and T. Michl. Abingdon, Oxon: Routledge.
- Mohun, Simon. 2014. "Unproductive Labor in the U.S. Economy 1964-2010." Review of Radical Political Economics, 46(3): 355–379.

- Mohun, Simon, and Roberto Veneziani. 2017. "Values, price and exploitation: the logic of the transformation problem." *Journal of Economic Surveys*, 31(5): 1387–1420.
- Mokre, Patrick, and Miriam Rehm. 2020. "Inter-industry wage inequality: persistent differences and turbulent equalisation." *Cambridge Journal of Economics*, 44(4): 99–942.
- Morishima, Michio. 1973. Marx's Economics. New York, NY: Cambridge University Press.
- Morishima, Michio. 1974. "Marx in the Light of Modern Economic Theory." *Econometrica*, 42(4): 611–632.
- Morishima, Michio, and F. Seton. 1961. "Aggregation in Leontief Matrices and the Labour Theory of Value." *Econometrica*, 29(2): 203–220.
- Morishima, Michio, and George Catephores. 1978. Value, Exploitation and Growth. London: McGraw Hill.
- Moseley, Fred. 1986. "Estimates of the Rate of Surplus-Value in the Postwar United States Economy." *Review of Radical Political Economics*, 18(1-2): 168–189.
- Moseley, Fred. 1988. "The Increase of Unproductive Labor in the Postwar U.S. Economy." *Review of Radical Political Economics*, 20(2-3): 100–106.
- Moseley, Fred. 1990. "The Decline of the Rate of Profit in the Postwar U.S. Economy: An Alternative Marxian Explanation." *Review of Radical Political Economics*, 22(2-3): 17–37.
- Moseley, Fred. 1991. The Falling Rate of Profit in the Postwar United States Economy. Basingstroke: Macmillan.
- Moseley, Fred. 1993. "Marx's Logical Method and the 'Transformation Problem'." In *Marx's Method* in *Capital.*, ed. F. Moseley. Atlantic Highlands, NJ: Humanities Press.
- Moseley, Fred. 1994. "Unproductive Labor and the Rate of Profit: A Reply to Cullenberg's Comment." *Review of Radical Political Economics*, 26(2): 121–128.
- Moseley, Fred. 1998. "Marx's Logic in Capital and the 'Transformation Problem'." In Marxian Economics: A Reappraisal., ed. R. Bellofiore. Basingstoke: Macmillan Press Ltd.
- Moseley, Fred. 2000. "The 'New Solution' to the Transformation Problem: A Sympathetic Critique." *Review of Radical Political Economics*, 32(2): 282–316.
- Ochoa, Eduardo M. 1989. "Values, prices, and wage-profit curves in the us economy." Cambridge Journal of Economics, 13(3): 413–429.
- **Paitaridis, Dimitris, and Lefteris Tsoulfidis.** 2012. "The Growth of Unproductive Activities, the Rate of Profit, and the Phase-Change of the U.S. Economy." *Review of Radical Political Economics*, 44(2): 213–233.
- Ricardo, David. 1951. On the Principles of Political Economy and Taxation. Cambridge, UK: Cambridge University Press. Reprinted in 2004 by Liberty Fund, Inc., Indianapolis, IN.
- Rieu, Dong-Min, and Hyun Woong Park. 2020. "Unproductive Activities and the Rate of Surplus Value at the Industry Level in Korea, 1995–2015." Journal of Contemporary Asia, 50(2): 284–307.
- Rieu, Dong-Min, and Hyun Woong Park. 2022. "Estimating the rate of surplus-value at the industry level: A case study of the South Korea using the inverse transformation method." *El trimestre económico*, 89(354): 567–586.
- Robinson, Joan. 1950. "Review of P. Sweezy, Karl Marx and the Close of his System by Eugen von Böhm-Bawek; Böhm-Bawerk's Criticism of Marx by Rudolph Hilferding; On the Correction of Marx's Fundamental Theoretical Construction in the Third Volume of Capital by Ladislaus von Bortkiewicz." *Economic Journal*, 60(238): 358–363.
- Robinson, Joan. 1964. Economic Philosophy. . 2nd ed., Harmondsworth: Penguin.
- Roemer, John E. 1982. A General Theory of Exploitation and Class. Cambridge, MA: Harvard University Press.
- Rubin, I. I. 1990. Essays on Marx's Theory of Value. Montréal, Québec: Black Rose Books. Translated by Miloš Samardžija and Fredy Perlman from the third edition, Moscow, Leningrad: Gosudarstvennoe Izdatel'stvo, 1928.
- Samuelson, Paul A. 1970. "The 'Transformation' from Marxian 'Value' to Competitive 'Prices': A Process of Rejection and Replacement." Proceedings of the National Academy of Sciences of the United States of America, 67(1): 423–425.
- Samuelson, Paul A. 1971. "Understanding the Marxian Notion of Exploitation: A Summary of the So-Called Transformation Problem Between Marxian Values and Competitive Prices." *Journal of*

Economic Literature, 9(2): 399–431.

- Samuelson, Paul A. 1974. "Insight and Detour in the Theory of Exploitation: A Reply to Baumol." Journal of Economic Literature, 12(1): 62–70.
- Scharfenaker, Ellis, and Duncan K. Foley. 2017. "Quantal Response Statistical Equilibrium in Economic Interactions: Theory and Estimation." *Entropy*, 19: 444–459.
- Scharfenaker, Ellis, and Gregor Semieniuk. 2017. "A Statistical Equilibrium Approach to the Distribution of Profit Rates." *Metroeconomica*, 68(3): 465–499.
- Sen, Amartya. 1978. "On the labour theory of value: some methodological issues." *Cambridge Journal* of *Economics*, 2(2): 175–190.
- Shaikh, Anwar. 1974. "Laws of Production and Laws of Algebra: The Humbug Production Function." *Review of Economics and Statistics*, 56(1): 115–120.
- Shaikh, Anwar. 1977. "Marx's Theory of Value and the 'Transformation Problem'." In *The Subtle Anatomy of Capitalism.*, ed. J. Schwartz. Culver City, CA: Goodyear Publishing Co.
- Shaikh, Anwar. 1978. "Political economy and capitalism: notes on Dobb's theory of crisis." Cambridge Journal of Economics, 2(2): 233–251.
- Shaikh, Anwar. 1979. "Notes on the Marxian Notion of Competition." Unpublished.
- Shaikh, Anwar. 1980a. "Laws of Production and Laws of Algebra—Humbug II." In Growth, Profits and Property., ed. E. J. Nell. Cambridge: Cambridge University Press.
- Shaikh, Anwar. 1980b. "Marxian competition versus perfect competition: further comments on the so-called choice of technique." *Cambridge Journal of Economics*, 4(1): 75–83.
- Shaikh, Anwar. 1981. "The Poverty of Algebra." In *The Value Controversy.*, ed. I. Steedman and P. Sweezy. London: New Left Books.
- Shaikh, Anwar. 1982. "Neo-Ricardian Economics: A Wealth of Algebra, A Poverty of Theory." Review of Radical Political Economics, 14(2): 67–83.
- Shaikh, Anwar. 1984. "The Transformation from Marx to Sraffa." In *Ricardo, Marx, Sraffa.*, ed. E. Mandel and A. Freeman. New York, NY: Verso.
- Shaikh, Anwar. 1992. "Values and Value Transfers: A Comment on Itoh." In *Radical Economics.*, ed. B. Roberts and S. Feiner. Boston, MA: Kluwer Academic Publishing.
- Shaikh, Anwar. 1998. "The Empirical Strength of the Labor Theory of Value." In Marxian Economics: A Reappraisal, Vol. 2., ed. R. Bellofiore. New York: St. Martin's Press.
- Shaikh, Anwar. 2008. "Competition and Industrial Rates of Return." In Issues in Finance and Industry., ed. P. Arestis and J. Eatwell. New York, NY: Palgrave Macmillan.
- Shaikh, Anwar. 2016. Capitalism: Competition, Conflict, Crises. New York, NY: Oxford University Press.
- Shaikh, Anwar, and E. Ahmet Tonak. 1994. *Measuring the Wealth of Nations*. New York, NY: Cambridge University Press.
- Smith, Adam. 2000. *The Wealth of Nations*. New York, NY: Random House, Inc. Modern Library Edition. Originally published in 1776 as: An Inquiry into the Nature and Causes of the Wealth of Nations.
- Steedman, Ian. 1977. Marx After Sraffa. London: New Left Books.
- Szumski, Jerzy S. 1989. "The transformation problem solved?" Cambridge Journal of Economics, 13(3): 431–452.
- **Torres-González, Luis Daniel.** 2022. "The Characteristics of the Productive Structure Behind the Empirical Regularities in Production Prices Curves." *Structural Change and Economic Dynamics*, 62: 622–659.
- Torres-González, Luis Daniel, and Jangho Yang. 2019. "The persistent statistical structure of the US input- output coefficient matrices: 1963-2007." *Economic Systems Research*, 31(4): 481–504.
- **Tsoulfidis, Lefteris, and Persefoni Tsaliki.** 2005. "Marxian Theory of Competition and the Concept of Regulating Capital." *Review of Radical Political Economics*, 37(1): 5–22.
- **Tsoulfidis, Lefteris, and Thanasis Maniatis.** 2002. "Values, prices of production and market prices: some more evidence from the Greek economy." *Cambridge Journal of Economics*, 26: 359–369.
- Veneziani, Roberto. 2012. "Analytical Marxism." Journal of Economic Surveys, 26: 649–673.
- Veneziani, Roberto, and Naoki Yoshihara. 2015. "Exploitation in economies with heterogeneous preferences, skills and assets: An axiomatic approach." Journal of Theoretical Politics, 27(1): 8–33.

- Veneziani, Roberto, and Naoki Yoshihara. 2017*a*. "Globalisation and Inequality in a Dynamic Economy: An Axiomatic Analysis of Unequal Exchange." *Social Choice and Welfare*, 49(3-4): 445–468.
- Veneziani, Roberto, and Naoki Yoshihara. 2017b. "One million miles to go: taking the axiomatic road to defining exploitation." *Cambridge Journal of Economics*, 41(6): 517–524.
- Wright, Ian. 2014. "A category-mistake in the classical labour theory of value." Erasmus Journal for Philosophy and Economics, 7(1): 27–55.
- Wright, Ian. 2019. "Marx's transformation problem and Pasinetti's vertically integrated subsystem." *Cambridge Journal of Economics*, 43: 169–186.
- **Yoshihara**, Naoki. 2021. "On the labor theory of value as the basis for the analysis of economic inequality in the capitalist economy." *The Japanese Political Economy*, 47(2-3): 190–212.
- Yoshihara, Naoki, and Roberto Veneziani. 2018. "The theory of exploitation as the unequal exchange of labour." *Economics and Philosophy*, 34(3): 381–409.
- Yoshihara, Naoki, and Roberto Veneziani. 2023. "The measurement of labour content: An axiomatic approach." Journal of Economic Behavior and Organization, 212: 392–402.